**Author's response to reviews**

**Title:** Cloxacillin versus Vancomycin for Presumed Late-Onset Sepsis in the Neonatal Intensive Care Unit and the Impact upon Outcome of Coagulase Negative Staphylococcal Bacteremia: A Retrospective Cohort Study

**Authors:**

- Sarah L Lawrence (SLawrence@cheo.on.ca)
- Virginia Roth (vroth@ottawahospital.on.ca)
- Robert Slinger (slinger@cheo.on.ca)
- Baldwin Toye (btoye@ottawahospital.on.ca)
- Isabelle Gaboury (igaboury@cheo.on.ca)
- Brigitte Lemyre (blemyre@ottawahospital.on.ca)

**Version:** 2  **Date:** 25 February 2005

**Author's response to reviews:**

To the editors of BioMed Central Pediatrics, Feb 25, 2005  
I have resubmitted my manuscript entitled Cloxacillin versus Vancomycin for Presumed Late-Onset Sepsis in the Neonatal Intensive Care Unit and the Impact upon Outcome of Coagulase Negative Staphylococcal Bacteremia: A Retrospective Cohort Study  
with the changes requested: I have added the title page, abstract, competing interests, authors contributions, acknowledgements, and references directly to the manuscript rather than as attached files. I have added a declaration of no competing interests to the manuscript. I have made no other changes. I hope this is satisfactory,  
Dr Sarah Lawrence